CENWP-OP-B                                                                                                                                  04 February 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 05 of 2003, which covers the period 26 January through 01 February 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 107.7 to 233.7 kcfs.  Daily project tailwater elevation averaged 11.4’ to 21.5’ msl.  Daily project forebay elevation averaged 74.8’ to 75.8’ msl.  River temperature averaged 40 to 41° F.  Daily turbidity measured 3.5’ to 7.0’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, are now closed 24 hours.  

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 16 remains out of service for an overhaul.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  During fish passage season, the units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Fish Unit 1 remains out of service for overhaul and stator rewind.   F2 was out of service from 1306 until 1800 to facilitate the Washington Shore collection channel inspection dive.

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed for cleaning.

2.1.1 FV1-1: 0855 until 1027 on 28 January.

2.2 STS/VBS Inspections.  No STS/VBSs were inspected this reporting week. 

2.3 Dewatering and Fish Salvage.  Unit 16 tail logs were pulled on 29 January.  Six sculpin and two channel catfish were released back to the river.

3.  RESEARCH.

3.1 NMFS Lamprey.  Tracking continues.

3.2 WDFW Fish Counting.  24-hr video counts at Bradford Island and Washington Shore.

4.  FISHWAYS 
4.1 Project biologists inspected 27-31 January this week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	Main dam, 

PH1, PH2
	Avian wires broken or removed
	Vandalism and contractor work.
	Will be repaired this winter.
	

	1/28-1/30
	PH1 PCC velocity
	Velocity <1.5 fps on electronic meter.
	Unknown
	No action taken.
	

	1/29 
	Weir 37
	Reading < 1.0’
	Unknown
	
	

	All week
	PH2 PCC velocity
	Velocity <1.5 fps.
	Meter is missing
	see misc. 5.5
	


4.3 Fish Ladder Temperatures.  Temperatures are not recorded in the winter.

4.4 Sensor Calibration.  Calibration was checked on 30 January this reporting week.

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Cascade Island controls are in manual mode awaiting tailwater sensor calibration.  Electricians are awaiting parts.  B-branch is also in manual mode, pending installation of a tailwater stillwell.

5.2 During low tailwater, the PH1 weir gates are unable to maintain greater than 8’ openings.

5.3 On 24 January, the Project tailwater hard constraint decreased to 11.0’ msl for all hours.  The soft constraint went to 11.2’ msl between 0700-1900 hours.  This is for the protection of the chum redds around Ives Island.

5.4 Because of  divers inspecting the collection channel, PH1 entrance gates were closed from 0936 until 1150 on 27 January.

5.5 Divers discovered the PH2 velocity meter was missing from the collection channel.  This explains why the velocity has been out of criteria.  The project has a spare meter to be installed during the mid season dives this summer.

5.6 DSM1 orifice lights timer has been disconnected.  DSM2 orifice lights timer will be disabled before the channel comes back in service.   Orifice lights will remain on 24 hours a day as per FPOM.  

5.7 Cascades Island returned to service at 1600 on 28 January.

5.8 A 1% violation was requested from 1500 until 2300 on 28 January.  This request was to facilitate corner collector dive operations by fully loading PH1 units before turning on units at PH2.

5.9 Bradford Island went to orifice flow on 01 February in preparation for the dewatering on 03 February.

 JAMES R. MAHAR

                                 

                                                           Operations Manager
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